Author's response to reviews

Title: Prevalence and associated factors of anemia among non-pregnant women of reproductive age in Bangladesh: Multilevel logistic regression analysis

Authors:

Md Kamruzzaman (kzaman_stat@yahoo.com)
Md Golam Rabbani (mgrstat57@gmail.com)
Saw Aik (sawaik@hot.com)
Md Abu Sayem (sayem072003@yahoo.com)
Md Golam Hossain (hossain95@yahoo.com)

Version: 3  
Date: 19 January 2015

Author's response to reviews:

Date: January 19, 2015
Dear Editor

We resubmitted the manuscript of our paper “Prevalence and associated factors of anemia among non-pregnant women of reproductive age in Bangladesh: Multilevel logistic regression analysis” (MS: 1785788281151373) to your Journal. According to your recommendation, language and grammar of our manuscript have been checked by a native English speaker. One of our Co-author wanted to put his name as Aik Saw instead of Saw Aik, we already put in revised manuscript. We hope our manuscript will be considered for publication. Please ask me if you find any problem in our manuscript. We are waiting for your positive response.

With best regards
Dr. Md. Golam Hossain
Professor of Health Research Group
Department of Statistics, University of Rajshahi
Rajshahi-6205, Bangladesh
E-mail: hossain95@yahoo.com
Tel. +880721750797 (Res.)